Author's response to reviews

Title: Integrated personal health records: Transformative tools for consumer-centric care

Authors:

Don E. Detmer (detmer@amia.org)
Meryl Bloomrosen (meryl@amia.org)
Brian Raymond (Brian.Raymond@kp.org)
Paul Tang (pctang@pacbell.net)

Version: 9 Date: 1 October 2008

Author's response to reviews:

We have corrected several typos.

1. Author affiliation: Don's email address has the character "I" after it that should be removed
2. Author affiliation: shouldn't Paul's title be Chief Medical Information Officer, not "Office"
3. Bullets: most of the bullets are square, but there are a few places where they are round: Last bullet on p. 6; first bullet on p. 12; all bullets on p. 19. Also, on p. 14, at the top, the heading "U.S. Federal Government Programs" should not be a bullet, but a run-in head same as "New Models..." on the same page.
4. Spacing: there are numerous places throughout where the spacing between paragraphs is inconsistent. I don't know if this is something to worry about. But it does look particularly bad when the inconsistency occurs within a bullet list like on the top set of bullets on p. 12; the bullets on p. 15; and the bullets on p. 18.
5. References:

   I found an error in one of the references: Number 27 should be The American College of Physicians: Patient-Centered Medical Home Overview. [http://www.acponline.org/advocacy/where_we_stand/medical_home/overview.htm]

   Reference 13: Need a space between "13." and the author
   Reference 22 (the one I gave you new info for in the last go-round of checking), there is a colon and a space missing after "Kim Y" Also, the first word in the title, Patient, should be bold.
   Reference 24: need a space between "24" and the author
   Reference 57 needs a period after June.
   Reference 69: Remove the comma after the word "commendations" at the end of the title.

6. Other:
On p. 2, end of last full paragraph: remove second period at end of sentence.
On p. 11, end of first bullet, period after the reference bracket looks like it is underlined.
On p. 17, 3rd paragraph from the bottom, last sentence, WebMd should be WebMD.